
diff -r 60f64793266f -r 2f4296a0ab21 handouts/ho03.tex
--- a/handouts/ho03.tex	Mon Oct 05 20:42:11 2015 +0100
+++ b/handouts/ho03.tex	Tue Oct 06 09:42:58 2015 +0100
@@ -186,17 +186,17 @@
 make \pcode{gcc} generate assembly instructions if you call it
 with the \pcode{-S} option, for example \pcode{gcc -S out
 in.c}\;. Or you can look at this code by using the debugger.
-How to do this will be explained later.}
+How to do this will be explained in the last section.}
 
 \begin{center}\small
-\begin{tabular}[t]{p{10cm}}
+\begin{tabular}[t]{p{11cm}}
 {\lstinputlisting[language={[x86masm]Assembler},
   morekeywords={movl},xleftmargin=5mm]
   {../progs/example1a.s}}
 \end{tabular}
 \end{center}
 \begin{center}\small
-\begin{tabular}[t]{p{10cm}}
+\begin{tabular}[t]{p{11cm}}
 {\lstinputlisting[language={[x86masm]Assembler},
   morekeywords={movl,movw},xleftmargin=5mm]
   {../progs/example1b.s}}  




